Author's response to reviews

Title: Development and evaluation of a food environment survey in three urban environments of Kunming, China

Authors:

Jenna Hua (jenna_hua@berkeley.edu)
Edmund Seto (seto@berkeley.edu)
Yan Li (yanli20021965@21cn.com)
May Wang (maywang@ucla.edu)

Version: 6 Date: 20 February 2014

Author's response to reviews:

Dear Editor,

This manuscript has been reviewed by several native English speakers, and necessary changes were made.

Thank you very much for your consideration.

Sincerely,

Jenna